\documentclass[11pt]{article}
\input{macros-course}

\begin{document}

\handout{student-handback}{Information Sheet}{September 9, 1999}

Please complete the information form below and
\begin{center}
{\large \bf return this sheet to us at the end of lecture today,}
\end{center}
so that we can complete your assignment to a tutorial section.

%\vskip 0.2in

\iffalse
In addition to using this information to determine tutorial assignments,
we will distribute a class list with phone numbers and email addresses 
to facilitate collaboration.  If you do not want your phone number or email 
address distributed by us, please say so below.
\fi

\bigskip

\def\lineitem#1{
\hbox to \hsize
  {\hbox{#1} \hrulefill }
\medskip}

\lineitem{Name}
\lineitem{Department and year}
\lineitem{Phone number}
\lineitem{Email address}

\medskip

Each student must attend a 1 1/2 hour tutorial section on Friday.  Write
numbers 1, 2 and 3 next to your first three choices of tutorial sections.
Write an X next to those tutorials which you cannot attend because of an
\emph{unavoidable} conflict.

\medskip
\begin{verbatim}
              Start     End               (Room)
             -------  -------   ------------------------
             9am      10:30am   ___(26-168)  ___(26-322)
             10am     11:30am   ___(2-142)   ___(36-372)
             10:30am  noon      ___(8-119)   ___(24-307)
             11am     12:30pm   ___(26-210)
             11:30am  1pm       ___(2-139)   ___(2-147)
             1pm      2:30pm    ___(26-168)  ___(36-839)
             2pm      3:30pm    ___(24-402)  ___(24-407)
             2:30pm   4pm       ___(26-310)  ___(36-144)
             3:30pm   5pm       ___(26-168)  
\end{verbatim}   

\bigskip
Please list all Course VI and Course XVIII subjects you have taken or 
are currently taking.  \textbf{Circle} those you are currently taking.

\begin{verbatim}

_______  _______  _______  _______  _______  _______  _______  _______ 

_______  _______  _______  _______  _______  _______  _______  _______ 
\end{verbatim}

We will post tutorial assignments on the course webpage
\begin{center}
\texttt{http://theory.lcs.mit.edu/classes/6.042/fall99}
\end{center}
as soon as possible, but probably not by tomorrow morning, so on Friday,
September 10, you may attend any convenient tutorial session.

\end{document}

